Author’s response to reviews

Title: The gene-treatment interaction of paraoxonase-1 gene polymorphism and statin therapy on insulin secretion in Japanese patients with type 2 diabetes: Fukuoka Diabetes Registry

Authors:

Akiko Sumi (cbjbx559@ant.bbiq.jp)
Udai Nakamura (uday@med.kyushu-u.ac.jp)
Masanori Iwase (iwase@intmed2.med.kyushu-u.ac.jp)
Hiroki Fujii (fujihi@intmed2.med.kyushu-u.ac.jp)
Toshiaki Ohkuma (ohkuma-t@intmed2.med.kyushu-u.ac.jp)
Hitoshi Ide (ide-kin@intmed2.med.kyushu-u.ac.jp)
Tamaki Jodai-Kitamura (tamakijodai@yahoo.co.jp)
Yuji Komorita (komo1113@gmail.com)
Masahito Yoshinari (yoshima@intmed2.med.kyushu-u.ac.jp)
Yoichiro Hirakawa (you1@eph.med.kyushu-u.ac.jp)
Atsushi Hirano (atsuhira@intmed2.med.kyushu-u.ac.jp)
Michiaki Kubo (michiaki.kubo@riken.jp)
Takanari Kitazono (kitazono@intmed2.med.kyushu-u.ac.jp)

Version: 2 Date: 30 Nov 2017

Author’s response to reviews:

Responses to the technical comments

1. Title page: Please include the email address of all authors.

Response: We include the email address of all authors in title page, page 1.

2. Abstract: Please change the header ‘Objective’ to ‘Background’ and ‘Research Design and Methods’ to ‘Methods’ in order to adhere to our submission guidelines.
Response: We change the header ‘Objective’ to ‘Background’ and ‘Research Design and Methods’ to ‘Methods’, page 4.

3. Abbreviations: Please move this section after Conclusions.
Response: We move this section after Conclusions.

4. Background: Please add this header at the beginning of the main manuscript text.
Response: We add this header at the beginning of the main manuscript text, page 7.

5. Research Design and Methods: Please change this header to ‘Methods’.
Response: We change this header to ‘Methods’, page 8.

6. Discussion: It came to our attention that your manuscript does not have a Discussion section. After careful assessment, we advise to change the ‘Conclusions’ to header to ‘Discussion’, and to add the ‘Conclusions’ header before the last paragraph of the main manuscript text.
Response: We change the ‘Conclusions’ to header to ‘Discussion’, and to add the ‘Conclusions’ header before the last paragraph of the main manuscript text, page 15-17.

Response: We write gene names in italic.

8. Declarations: Please note that all manuscripts must all the following sections under heading ‘Declarations’. The declarations should follow the abbreviations section, and be before the References.

Ethics approval and consent to participate

Consent for publication
Availability data and material

Competing interests

Funding

Author’s contributions

Acknowledgements

Response: We contain the above sections under the heading ‘Declarations, page 17-19.

9. We include ‘Additional files’ section in accordance with BioMed Central editorial policies and formatting guidelines.